Author's response to reviews

Title: Prediction of depression in European general practice attendees: the PREDICT study

Authors:

Michael King (m.king@medsch.cul.ac.uk)
Scott Weich (S.Weich@warwick.ac.uk)
Francisco Torres (ff Torres@urg.es)
Igor Svab (igor.svab@mf.uni-lj.si)
Heidi Maaroos (Heidi.Ingrid.Maaroos@ut.ee)
Jan Neeleman (J.Neeleman@umcutrecht.nl)
Miguel Xavier (xavierm@sapo.pt)
Richard Morris (r.morris@pcps.ucl.ac.uk)
Carl Walker (c.walker@medsch.ucl.ac.uk)
Juan A Bellon (jabellon@terra.es)
Berta Moreno (bertamk@uger.es)
Danica Rotar (Danica.rotar@quest.arnes.si)
Janez Rifel (janez.rifel@MF.UNI-LJ.SI)
Anu Aluoja (Anu.Aluoja@klinikum.ee)
Ruth Kalda (Ruth.Kalda@ut.ee)
Mirjam I Geerlings (M.Geerlings@umcutrecht.nl)
Idalmiro Carraca (xavierm@sapo.pt)
Manuel Caldas de Almeida (jcaldasalmeida@aol.com)
Benjamin Vincente (bvicente@udec.cl)
Sandra Saldivi (ssaldivi@udec.cl)
Pedro Riosco (bvicente@udec.cl)
Irwin Nazareth (.nazareth@pcpl.ucl.ac.uk)

Version: 4 Date: 25 November 2005

Author's response to reviews:

Following your email, please find attached a version of the paper with the formatting changes carried out as requested.

Title & Abstract are identical in the manuscript and in the submission system.

Author list - The list of authors in the manuscript are written exactly as they are in the submission system.

Qualifications - removed the authors’ qualifications (e.g. PhD, M.D.) from the title page.

Affiliations - affiliations be written in full

each author linked to their corresponding affiliation with the use of superscript numbers.

Section headings - sentence case used for all headings and sub-headings in the manuscript

Introduction - renamed 'Background'.

References - References cited in the text using consecutive normal case numbers in square brackets

References - In the manuscript text, the reference are placed before the full stop.

DESIGN AND METHODS - changed to Methods/design.

Manuscript sections - Manuscript sections include (in the following order): Abstract; Background; Methods/Design; Discussion; Abbreviations (if any); Competing interests; Authors’ contributions; Acknowledgements; References; Figure legends (if any); Tables (if any); Description of Additional files (if any).
Author's contributions - changed to Authors' contributions.

Tables - tables moved to after the references.